Author's response to reviews

Title: Entry quarantine and airport sentinel surveillance for dengue infection in Taiwan

Authors:

Mei-Mei Kuan (kuan@cdc.gov.tw)
Feng-Yee Chang (fychang@cdc.gov.tw)

Version: 3 Date: 25 April 2011

Author's response to reviews:

Dear Dr. Diana Marshall and editors,

Appreciate your e-mail and your suggestion for improving our manuscript. We have processed it with the following points:

1) Ensured that the advance of the current submission over our previous work was made clear in the manuscript, and ensured that there was no overlap in the data presented in the two submissions.

2) Revised the Methods section to make the methodology used in the study clearer.

3) We have ever used a professional copyediting service for editing the English.

4) The tables were revised the formatting for clarity.

5) Manuscript sections – Ensured that the manuscript sections were in the correct order including: Abstract; Background; Methods; Results; Discussion; Conclusions; Competing interests; Authors' contributions; Acknowledgements; References; Figure legends; Tables;

Best wishes,

Mei-Mei
Mei-Mei Kuan, PhD
email: kuan@cdc.gov.tw